Point-by point response to the Reviewers’ comments

Reviewer 1.
Thank you very much for the kind words of appreciation of our paper. Concerning the comments:
  1. A table for clinical and lab presentation of all patients, along with course during hospital stay was made and included as Table 2.
  2. We included an Abstract with key words.
  3. A Patient’s consent statement was included as requested.

Reviewer 2.
Thank you very much for the kind words of appreciation of our paper.